| <sup>12</sup> 64 |  |
|------------------|--|
| UT UT            |  |

## medio ambiente y desarrollo

A systems approach to sustainability and sustainable development

Gilberto Gallopín





Sustainable Development and Human Settlements Division

ECLAC/ Government of the Netherlands Project NET/00/063 "Sustainability Assessment in Latin America and the Caribbean"

Santiago, Chile, March, 2003

This document was prepared by Mr. Gilberto Gallopín, Regional Adviser on Environmental Policy of the Sustainable Development and Human Settlements Division of ECLAC in the framework of the Project NET/00/063 sponsored by the Government of the Netherlands. The author is very grateful to Jean Acquatella, Alicia Bárcena, Isabel A. Gómez, Carlos de Miguel and Andrés Schuschny for their comments on drafts of this paper.

This document has been reproduced without formal editing. The views expressed herein are those of the author and do not necessarily reflect the views of the Organization.

United Nations Publication LC/L.1864-P ISBN: 92-1-121398-3 ISSN printed version: 1564-4189 ISSN electronic version: 1680-8886 Copyright © United Nations, March 2003. All rights reserved Sales number: E.03.II.G.35 Printed in United Nations, Santiago de Chile

Application to the right to reproduce this work are welcome and should be sent to the Secretary of the Publication Board, United Nations Headquarters, New York, N.Y. 10017, USA. Member States and their Governmental Institutions may reproduce this work without prior authorization, but are requested to mention the source and inform the United Nations of such reproduction.

## Index

| Abs  | stra | ct  | 5        |
|------|------|---|----------|
| Ι.   | Int  | roduction   | 7        |
| II.  | Ab   | basic notion of sustainability                            | 9        |
| III. | Th   | e subject of sustainability                               | 13       |
| IV.  | Lo   | oking for fundamental attributes underlying               |          |
|      | SU   | stainability at the level of whole socio-ecological stems | 17       |
| V.   | Su   | stainable development                                     | 17<br>19 |
|      | a)   | The ethical foundations of sustainable development        | 20       |
|      | b)   | Dynamism  | 21       |
|      | c)   | Concept   | 21       |
|      | d)   | Implementation  | 22       |
|      | e)   | A variety of perspectives                                 | 23       |
| VI.  | Th   | e different guises of development                         | 25       |
| VII. | Alt  | ernative paradigms for sustainable development .          | 29       |
|      | a)   | Paradigm 1. Standardization                               | 31       |
|      | b)   | Paradigm 2. Optimization                                  | 32       |
|      | c)   | Paradigm 3. Pessimization                                 | 32       |
|      | d)   | Paradigm 4. Equitization                                  | 32       |
|      | e)   | Paradigm 5. Stabilization                                 | 33       |
| VIII | . Ćo | nclusions   | 35       |
| Ref  | ere  | nces  | 37       |
| Ser  | ie N | ledio ambiente y desarrollo: published numbers            | 39       |

## Index of tables

|  | Table 1 | Theoretical perspectives | on sustainable development |  |
|--|---------|--------------------------|----------------------------|--|
|--|---------|--------------------------|----------------------------|--|

## Index of figures

| Figure 1:  | An open system; the state variables are those internal to the system          | . 10 |
|------------|---|------|
| Figure 2:  | State transitions of a finite-state system                                    | . 11 |
| Figure 3:  | The extreme anthropocentric position  | . 14 |
| Figure 4:  | The extreme biocentric position   | . 14 |
| Figure 5:  | A socio-ecological system   | . 15 |
| Figure 6:  | A socio-ecological system representation emphasizing inclusion relations      | . 16 |
| Figure 7:  | A socio-ecological system (large oval) representation emphasizing functional  |      |
|            | relations, internal and with the external world                               | . 16 |
| Figure 8:  | A set-theoretical representation of development, sustainability, and economic |      |
|            | growth, and quality of life (QOFL)  | . 26 |
| Figure 9:  | The different guises of development   | . 27 |
| Figure 10: | Alternative development paths   | . 28 |
| Figure 11: | An idealized representation of the state space of a socio-ecological system   | . 30 |
| Figure 12: | A generic state space for representing the coevolution of N and H             | . 31 |
| Figure 13: | State trajectories originating from the current state P <sub>0</sub>          | . 31 |

## Abstract

The concepts of sustainability and sustainable development are analyzed from a systems perspective. In the most general terms, sustainability of any system can be represented by a non-decreasing valuation function of the outputs of interest of the system considered.

Different perspectives on the system of reference are discussed, from the extreme anthropocentric to the extreme bio- or ecocentric positions, and related to the criteria (based on the assumed substitutability between natural and manufactured capital) of very strong, strong, weak, and very weak sustainability.

A set of underlying determinants of sustainability is proposed and discussed, including availability of resources, adaptability/ flexibility, homeostasis, capacity of response, self-reliance, and empowerment.

The concept of sustainable development is discussed and alternative theoretical perspectives that have been used in the literature are presented.

The relationship between sustainability, development, nondevelopment, and maldevelopment; and material and non-material economic growth is mapped as a Venn diagram; alternative trajectories towards sustainable development for rich and poor countries are identified.

Five alternative paradigms/strategies for sustainable development are summarized, showing the complexity of the process of choosing the right actions to move towards sustainable development.

### I. Introduction

The concept of sustainability and particularly of sustainable development figure among the most ambiguous and controversial in the literature. The present document is an attempt to examine these concepts from a systems perspective, seeking to abstract their fundamental elements.

It is becoming increasingly clear that the quest for sustainability and sustainable development requires integrating economic, social, cultural, political, and ecological factors (UNCED 1992, Gallopín et al. 2001, Kates et al. 2001). It requires the constructive articulation of the top-down approaches to development with the bottom-up or grassroots initiatives. It requires the simultaneous consideration of the local and the global dimensions and of the way they interact. And it requires broadening the spatial and temporal horizons to accommodate the need for intra-generational as well as inter-generational equity.

In dealing with these issues, the systems approach can offer a perspective more useful than other analytical approaches, because the systems view is a way of thinking in terms of connectedness, relationships, and context.

Section II proposes a general definition of sustainability applicable to any open system, and distinguishing between sustainability of the outputs of the system and of the system itself. Section III addresses the important ideological differences associated with the choice of the system of reference in the discussions of sustainability, and the related economic concepts of weak and strong sustainability. A set of fundamental whole-system attributes underlying system sustainability is proposed in section IV. Section V introduces the concept of sustainable development and reviews the different theoretical perspectives from which it has been approached. In Section VI a discussion of different development situations is presented, distinguishing between development, underdevelopment, and maldevelopment in terms of their sustainability, desirability, and between the situation of rich and poor countries. Section VII discusses five alternative paradigms of sustainable development, which illuminate differences in goals and worldviews of strategic significance.

Finally, Section VIII presents a number of conclusions derived from the analysis performed.

## II. A basic notion of sustainability

The concept of sustainability is a complex one; however, it is possible to distil some of its most basic and general characteristics by adopting a systemic approach.

For the present purposes, a system is simply defined as a set of interrelated elements (or subsystems). The elements can be molecules, organisms, machines or their parts, social entities, or even abstract concepts. The relations, interlinkages, or "couplings"<sup>1</sup> between the elements may also have very different manifestations (economic transactions, flows of matter or energy, causal linkages, control pathways, etc.).

All physically existent systems are *open*, having exchanges of energy, matter and information with their environment that are significant for their functioning.<sup>2</sup> Therefore, what the system "does", its behavior, depends not only on the system itself, but also on the factors, elements or variables coming from the environment of the system and impinging on it (the "input variables"); on the other hand, the system generates variables that exert on the environment (the "output variables") as illustrated in Figure 1.

<sup>&</sup>lt;sup>1</sup> In abstract terms, the elements and the relation between the elements defines a system. The term "relation" is used here broadly to include also similar terms such as "constraint", "structure", "organization", "cohesion", "interaction", "interconnection", "correlation", "pattern".

<sup>&</sup>lt;sup>2</sup> Systems open to energy but closed to exchanges of matter are sometimes called *isolated*, but this distinction is not fundamental here.

# SYSTEM output variables

#### Figure 1 AN OPEN SYSTEM; THE STATE VARIABLES ARE THOSE INTERNAL TO THE SYSTEM

Source: Author's elaboration.

Thus, the state of the system (the set of values adopted by all the internal variables of the system)<sup>3</sup> at a given time, is determined by the previous state of the system and by the inputs received by the system in the last period of time<sup>4</sup>.

input variables

This can be represented (Gallopín, 1996), for notational simplicity (the same reasoning applies for continuous systems), by the canonical definition of a finite-state general system (Gill 1969):



Where S denotes the internal state of the system, I is the input vector (the list of all input variables) to the system, O is the output vector from the system, and F and G are functions (deterministic or probabilistic). The subindex t stands for time. The output variables are those considered of interest for the performance of the system; some (or all) of them may be state variables. In the general case, all the variables may vary over time, space and "population".<sup>5</sup> The pair of equations above defines the behavior of the system. See Figure 2 for a graphical representation.

In its most general definition, a state is "any well-defined condition that can be recognized if it occurs again" (Ashby, 1956).

<sup>4</sup> In general, the memory of past changes is imbedded in the current value of the state, and therefore this generic formulation also applies to systems with time-lags.

<sup>5</sup> Any meaningful grouping, such as Company, type, country, or income bracket.



## Figure 2 STATE TRANSITIONS OF A FINITE-STATE SYSTEM

Source: Author's elaboration.

Using this framework, sustainability can be defined in elementary terms<sup>6</sup> by:



Where V is a valuation function of the outputs of the system (i.e., a sustainable system is a system for which the net "worth" –not necessarily in economic terms– of the output produced is non-decreasing in time). Any valuation implies a strong subjective component, and therefore the specification of the function V (and the choice of output variables of interest) may vary widely reflecting the range of perceptions and positions regarding the relationships between nature and society (some of those views are discussed in the next section). For some, O is simply the total capital stock, and V is a monetary measure of that stock. Others may define V as some kind of aggregate welfare function, and O may be differentiated between natural, manufactured, and social capital. Or V could be a valuation function including some ethical priorities for the conservation of all living species, expressed in non-monetary units. It is precisely in the explicit or implicit constructive specification of the function and its arguments where many of the discrepancies about the meaning of sustainability and sustainable development become manifest.

Sometimes we are interested in the *sustainability of the system itself* (e.g., the preservation of a natural ecosystem such as a pristine forest); in this case, the output variables are the same as the state variables (in other words, what is sought is the preservation of the system itself). When the output variables are different from the state variables, we are referring to the *sustainability of the output(s) of the system* (e.g. agricultural yield of an agroecosystem<sup>7</sup>), not necessarily to the sustainability of the system.

When referring to the sustainability of a system it should be made clear which sustainability is being considered, because the implications may be quite different for each case. Sometimes, we want to sustain part of the output but change the system.<sup>8</sup> Sustainable development implies change; sometimes we want to improve or transform the system, sometimes we want to change the system to improve some of its outputs.

<sup>&</sup>lt;sup>6</sup> More sophisticated definitions are possible, such as using the integral of the outputs through time.

This can be related to the concept of "non decreasing production (or consumption) stream or flow" used by economists.

<sup>&</sup>lt;sup>8</sup> For example, when the intent is to move from a military dictatorship to a democratic system while simultaneously maintaining the manufacturing sector developed under the military dictatorship.

## III. The subject of sustainability

The preceding discussion helps to clarify some of the underlying differences in the debates on sustainability and sustainable development. Those who will only pay attention to the sustainability of the social, or socio-economic system,<sup>9</sup> and those who will privilege only the sustainability of Nature represent the poles. The alternative views can be characterized, in a simplified way, as follows:

Sustainability of the human system only. This position, if taken to the extreme, could result in the Earth becoming a totally artificialized planet if total substitutability of natural resources and services were possible. The classical economicist view, for instance, regards the economy as the relevant system, and relegates nature to the role of provider of natural resources and services and of a sink for the wastes produced by human activities (Figure 3). This is consistent with the notion of "very weak sustainability"<sup>10</sup> (Turner 1993). The very weak sustainability approach asserts that natural and manufactured capital can substitute perfectly for one another. The substitutability of different types of capital implies that the preservation of an aggregate level of natural plus manufactured capital, rather than the preservation of natural capital in particular, is crucial.<sup>11</sup> The sustainability of ecological systems is viewed as important only as far as required for the sustainability of the human component. But there is too much we do not know (in this situation

<sup>&</sup>lt;sup>9</sup> Note that the term "socio" includes all that is human (economic, social, demographic, cultural, etc.).

<sup>&</sup>lt;sup>10</sup> Pearce and Atkinson (1992) coined the concepts of weak and strong sustainability. Turner (1993) further subdivided them into very weak, weak, strong and very strong categories.

<sup>&</sup>lt;sup>11</sup> It includes the so-called "Hartwick-Solow sustainability" which requires maintenance of the total capital stock (natural and humanmade) of society, and "Hicksian sustainability" which requires non-decreasing comsumption-including consumption of environmental goods and services (Ayres et al. 1998).

the *precautionary principle*<sup>12</sup> is called for). Precautionary approaches are necessary in order to incorporate an appropriate level of risk aversion in the face of uncertainty. There is also the issue of desirability or preferences: would we like to live in an artificial planet?



## Figure 3 THE EXTREME ANTHROPOCENTRIC POSITION

Source: Author's elaboration.

## Figure 4 THE EXTREME BIOCENTRIC POSITION



Source: Author's elaboration.

**Sustainability of the ecological system** primarily, even if it means elimination or displacement of the human component (Figure 4). Those who would value ecological sustainability above and beyond, rather than equal or subordinate to, economic and social sustainability represent an extreme "deep green" position in opposition to the anthropocentric one. This perspective is consistent with the concept of "very strong sustainability". The very strong sustainability position asserts that natural resources cannot be substituted by human-made capital; they cannot be

<sup>&</sup>lt;sup>12</sup> Defined by the United Nations Rio Declaration on Environment and Development, 1992, in the following way: 'Where there are threats of serious or irreversible damage, lack of full scientific certainty shall not be used as a reason for postponing cost-effective measures to prevent environmental degradation.'

depleted, therefore, without an irreversible loss in social welfare. Very strong sustainability favors a more fundamentalist mode of ecological solidarity with the Earth and all forms of life. This view is most compatible with a steady-state economy. Here, the preservation of the environment –a biocentric viewpoint– is the ethical precondition for sustainability. Pursuing ecological sustainability by way of diminishing social and economic concerns, even to the point of excluding humans or increasing human poverty, is not acceptable for the majority of us.<sup>13</sup>



Source: Modified from Castri (1981).

**Sustainability of the whole socio–ecological system**. The only option that makes sense in the long-term is to seek the sustainability of the whole socio–ecological system. The rationale for considering the whole system is based upon the existence of important interlinkages between society and nature. A *socio–ecological system* (Gallopín *et al.*, 1989) is defined as any system composed of a societal (or human) component (subsystem) in interaction with an ecological (or biophysical) component. It can be either urban or rural, and it may be defined at different scales from local to global.<sup>14</sup> (See Figure 5 for an allegoric representation; Figures 6 and 7 are alternative systemic representations). This outlook is consistent with the notion of "strong sustainability". The strong sustainability approach holds that different types of capital are not necessarily substitutable, so that minimum amounts of a number of different types of capital (economic, ecological, social) should be independently maintained, in real physical/biological terms. The major motivation for this insistence is derived from the recognition that natural resources are essential inputs for economic production, consumption or welfare that cannot be substituted for by physical or human capital. It is understood that some environmental components are unique and that some

Figure 5

<sup>&</sup>lt;sup>13</sup> There are those who would hold that this position could be well-justified in some very specific, localized situations such as keeping people out of national parks; there is room for debate on this issue.

<sup>&</sup>lt;sup>14</sup> Local may be a household and its interactions with its immediate surroundings, and global is understood as the whole of humankind and its interactions with the natural world or biosphere.

environmental processes may be irreversible (over relevant time horizons). Therefore, strong sustainability implies that the aggregate amount of natural capital has to be maintained essentially at the present level. Under this notion, any development path that leads to an overall reduction of the stocks of natural capital (or, specially, to a decline below the minimum) fails to be sustainable even if other forms of capital increase.<sup>15</sup>



Source: Author's elaboration.

Source: Author's elaboration.

The sustainability of the whole socio-ecological system may also be compatible with the notion of "weak sustainability". Weak sustainability places emphasis on the value of safeguarding ecological and biogeochemical processes that are irrecoverable if lost. These processes and their associated species mix are referred to as *critical natural capital*. Critical natural capital should not be allowed to be substituted for, but otherwise manufactured capital of equal value can take the place of natural capital.<sup>16</sup> One major problem lies in the choice of criteria for assigning value to the ecological assets, considering the arguments about the incommensurability<sup>17</sup> of ecological and manufactured capital.

<sup>16</sup> A society is said to be weakly sustainable if well-being is non-declining from generation to generation.

<sup>&</sup>lt;sup>15</sup> Sustainability here is viewed as non-diminishing life opportunities (Ayres et al. 1998).

 <sup>&</sup>lt;sup>17</sup> Incapable of being measured against a common standard.

## IV. Looking for fundamental attributes underlying sustainability at the level of whole socio-ecological systems

What is required for a socio-ecological system to be sustainable? Different attributes may be identified, depending on the specific socio-ecological system and particular component (soil, vegetation, social group, etc.) considered. However, it seems reasonable that some generic systemic properties would be universally required for the sustainability of socio-ecological systems. A proposal for such a set of basic, total-system attributes has been made by Gallopín (1994). Some of these properties arise from the ecological and human subsystems, and others only from the human subsystem, but all are important at the whole socio-ecological system level. Those basic properties are:

- $_{x}$ Availability of Resources. This is an obvious attribute, and it can include *resources* (e.g. water, light, money, etc.) *assets* and *entitlements*.
- $_{\rm X}$ Adaptability and Flexibility (as opposed to Rigidity). A degree of plasticity<sup>18</sup> is required to detect and make sense of changes occurring in the outside world. If that capacity is lost, the system may become rigid and unable to detect changes. As the environment keeps changing undetected by the system, or detected

<sup>&</sup>lt;sup>18</sup> Plasticity being understood as the capacity of the whole socio-ecological system to be influenced and modified by its environment.

but without the system adapting to the new conditions, the system will eventually collapse, because its behavior will no longer be compatible with the new situation.

**General Homeostasis: Stability, Resilience, Robustness (as opposed to Vulnerability, Fragility).** This refers to the capacity of the system to maintain or preserve the values of essential variables around (near) a given trajectory or state (stability), a given domain of attraction (resilience), or a given system structure (robustness).

This capacity can erode slowly in a manner difficult to perceive. There are many examples in natural resources management in which this has happened, leading to loss of homeostasis at different levels. Exemplifying this, the rapid expansion of the cholera epidemics in Latin America in 1994 can be connected with the slow erosion of basic sanitation services during the 80's (the "lost decade" for the region). An important point (Nicolis and Prigogine 1977, Prigogine and Stengers 1979) is that the new structure arising from structural change is inherently unpredictable.

**Capacity of Response.** This refers to the capacity of a socio-ecological system to cope with change.

Capacity of response is in some way related to the capacity to maintain or enlarge the system's repertoire of options. It is also related to the "ability to switch strategies according to condition".

Capacity of response is built upon adaptability, homeostasis and awareness.

Self-reliance (as opposed to Dependency).<sup>19</sup> Self-reliance refers to the capacity of a socioecological system to regulate its interactions with its environment. Such a capacity depends upon the degree to which the system exercises control over its own interactions with its environment.

Empowerment. This attribute denotes the capacity of the socio-ecological system not just to respond to change, but to innovate and induce change in other systems in pursuit of its own goals. It should be noted that this attribute can apply specifically to the human subsystem, but not to the ecological subsystem.<sup>20</sup>

The above set of attributes represents a preliminary proposal, pointing to the need to think about basic underlying attributes for sustainability, rather than about attributes or properties of parts or components.

<sup>&</sup>lt;sup>19</sup> Self-reliance should not be confused with self-sufficiency nor with autarky.

<sup>&</sup>lt;sup>20</sup> In some sense, the process of natural evolution often led to species influencing other species (i.e. competition or mutualism) but this non-conscious "empowerment" unfolds at time scales much slower than the human actions considered here.

## V. Sustainable development<sup>21</sup>

Sustainability is not the same as constancy. Although sometimes sustainability is presented as meaning the maintenance of a fixed state of a system, this is not scientifically correct; even pristine ecosystems are in permanent change, involving renewal and destruction of components, adapting to changes in their environment and coevolving with it. There are many examples involving fisheries, managed forests and wildlife, and other forms of management of ecological resources that have shown that attempts to "freeze" the variables of the system in attempts to obtain "optimal performance" often lead to loss of resilience of the system and even its collapse (Holling, 1973, 1986).

All living systems are changing systems and the essential point is not to eliminate change, but to avoid the destruction of the sources of renewal,<sup>22</sup> from which the system can recover from the unavoidable stresses and disturbances to which it is exposed because of its condition of being an open system.

The concept of *sustainable development* is quite different from that of *sustainability*<sup>23</sup> in that the word "development" clearly points to the idea of change, of directional and progressive change. As will be discussed later, development does not necessarily mean quantitative growth, being more akin to the notion of qualitative unfolding of potentialities and increasing complexity (which, depending on the concrete situation, may or may not include or require quantitative growth).

<sup>&</sup>lt;sup>21</sup> From Gallopín and Christianson (2000).

<sup>&</sup>lt;sup>22</sup> The sources of renewal are often system-specific; for instance, the renewal of moist tropical forests depends critically on the maintenance of the understory, and the sources of renewal of many societal systems lay in the social and natural capital.
<sup>23</sup> Which can be applied to the maintenance of an ariting situation or system state.

<sup>&</sup>lt;sup>23</sup> Which can be applied to the maintenance of an existing situation or system state.

Here, what is sustained, or has to be made sustainable, is the process of improvement of the human condition (or better, of the socio-ecological system to which humans pertain), a process that does not necessarily require indefinite growth in the consumption of energy and materials.

We are living through a period of tremendous demographic, technological, and economic transformation. In an attempt to ensure that the changes affecting humanity are changes for the better, the world community has initiated the process of redefining progress. This attempt at redefining progress is known as sustainable development.

The speed and magnitude of global change, the increasing connectedness of social and natural systems, and the growing complexity of societies and of their impacts upon the biosphere, highlight that sustainable development must aim not only to preserve and maintain the ecological base for development and habitability, but also to increase the social and ecological capacity to cope with change, and the ability to retain and enlarge the available options to face a natural and social world in permanent transformation.

Thus, the concept of sustainable development cannot mean merely perpetuation of the existing situation. The central question is what is to be sustained, and what is to be changed. Moving towards sustainable development requires:

- x Removing accumulated rigidities and impediments;
- $_{\rm X}$  Identifying and protecting the accumulated foundations of knowledge and experience that are important as a basis upon which to build;
- $_{\rm X}$  Sustaining the social and natural foundations for adaptation and renewal, and identifying and enhancing the lost renewal capacity needed;
- x Stimulating innovation, experimentation and social creativity.

The body of literature pertaining to the topic of sustainable development is both voluminous and dissonant. The multitude of opinions on sustainable development may be indicative of the high-stakes involved. Indeed, trying to formulate a new framework for human reasoning capable of underlying the arrangements of our evolving society is of significant concern.

Amid the diversity of approaches to sustainable development there are, of course, some recurrent elements that provide some degree of internal consistency to the body of literature. The aim of this section is to review basic elements of the sustainable development concept.

#### a) The ethical foundations of sustainable development

Opinions about what constitutes the ethical foundation of sustainable development vary to some degree. One ethical concern that is often referred to in the sustainable development literature is that of intergenerational justice<sup>24</sup> (Costanza, 1991, Vercelli, 1998). This concern is explicitly mentioned in the general definition of sustainable development prepared by the World Commission on Environment and Development (see the following concept subsection for definition).

Notably, the idea that future generations should be compensated for reductions in the endowments of resources brought about by the actions of present generations is sometimes in tension with another often cited ethical basis for sustainable development –intragenerational equity–. Intragenerational equity is concerned with the reduction of resource disparities among those presently living today.

Another ethical concern, that complements the anthropocentric objectives of intergenerational and intragenerational justice, is the ecocentric concern for biodiversity,

<sup>&</sup>lt;sup>24</sup> Closely related to the "equitization" paradigm of sustainable development (see Section VI).

sometimes referred to as Biophilia (Wilson, 1986) and already visited in Section III. Biophilia represents an appreciation of the intrinsic values in nature. The ecocentric objective of Biophilia is to preserve diversity, from species to ecosystems (Bergh and Jeroen, 1996). Variations on Biophilia have been put forth by the Deep Ecology school of thought (Drengson and Inoue, 1995).

#### b) Dynamism

Sustainable development cannot exist as some static equilibrium state that can be regulated by reference to constant limits and some simple notion of balance between the various dimensions (Brooks, 1992). Permanent technological innovation and changes in social organization make sustainable development to be a dynamic process. Rates of change are important determinants of sustainable development (Froger and Zyla, 1998). A dedication to learning how rates of change affect the behavior of social, ecological and economic systems over time is an important part of the process of enabling sustainable development.

#### c) Concept

Since its introduction in the late 1970s the concept of sustainable development has suggested a synthesis between economic development and environmental preservation (Bergh and Jeroen, 1996). The need for this type of synthesis derives in large part from the fact that permanently decreasing environmental stocks cannot support increasing or perhaps even constant levels of material economic throughputs for an indefinite period of time (Drummond and Marsden, 1999).

Definitions of sustainable development hold in common a respect for the need to integrate economic and environmental concerns. Beyond this basic consideration, commonalties among definitions of sustainable development are more subtle.

The most often cited definition of sustainable development is the one proposed by the United Nations Commission on Environment and Development (otherwise known as the Brundtland Commission) in 1987 (WCED 1987). In its report to the United Nations General Assembly, entitled Our Common Future, the Commission defined sustainable development as "development that meets the needs of the present without compromising the ability of future generations to meet their own needs".

Many attempts to promote sustainable development involve strategies designed to define and subsequently monitor some form of "sustainability limits" (Farrell and Hart, 1998). This approach is premised upon the observation that natural resources are finite and that there are limits to the carrying capacity of the Earth's ecosystems.

Another approach to sustainable development, which is often linked in some manner with the sustainability limits approach, is the "competing objectives" approach, which focuses on reconciling social, economic and ecological goals (Peterson, 1997).

To generalize, approaches to sustainable development from an ecological perspective stress the importance of focusing on the societal ability to resist or recover from disturbances, stresses and shocks rather than on its ability to produce goods (Vercelli, 1998). Alternatively, approaches to sustainable development from an economic perspective define economic development (sometimes assumed to be synonymous with economic growth) as sustainable whenever a certain crucial variable may be 'sustained', in the sense that it is not bound to diminish in the future as a consequence of growth itself. In the literature, approaches to sustainable development may be grouped into three sets according to whether the crucial variable or objective function to be maximized is welfare (or utility), consumption, or (man-made and/or natural) capital. The choice of the crucial variable has far-reaching implications since its sustainability often implies the unsustainability of other plausible candidates for this role (Vercelli, 1998). The later author departed from those categorizations by proposing that the freedom of future generations should be taken as the basic variable to be preserved through time by economic development.

#### d) Implementation

If sustainable development is to be achieved, understanding the interlinkages between social, ecological and economic dimensions of our world is of significant importance. This is so because the behavior of a system is in general determined as much by the causal interlinkages between its variables than by the changes in the values of the variables themselves. To understand such interlinkages, it is worthwhile to adopt a systems approach to observing worldly phenomenon. A key feature of the systems approach is the recognition that outcomes are not necessarily predictable since our activities may 'force' a system into a whole new form of behavior (that could include collapse) never seen before (Holling, 1973, 1986; Gunderson *et al.*, 1995, IGBP 2001).

Complementary to the need for a systems approach is the need to integrate multiple perspectives into the process of implementing sustainable development. A noteworthy point regarding sustainable development, a point that differentiates the concept from narrower ideas such as environmentalism, is that the concept is more than the sum of its parts (Brooks, 1992).

A means of assessing progress towards sustainable development is integral to implementing the concept. Traditional market indicators are unable to signal whether or not the integrity of a natural system is being dangerously eroded. Thus these indicators need to be complemented by sustainability indicators. Some sustainability indicators that have been proposed refer to: reducing the impact that human activities have on the environment (particularly the rates at which renewable and nonrenewable resources are used); not exceeding the carrying capacity of natural resources and ecosystems; integrating long-term economic, social and environmental goals, and preserving biological, cultural and economic diversity (Bergh and Jeroen, 1996).

Sustainability indicators must ultimately be linked with achievable goals. Setting goals for sustainable development amid large groups of stakeholders is a most difficult process. With large groups of stakeholders, diversity of value perspectives, derived from different life experiences and cultural histories, tends to undermine the possibility of any consensus on the criteria for sustainable development (Peterson, 1997). This is particularly relevant to the global scale. To achieve global sustainable development, in light of this constraint, it may be best to allow for different regions to characterize sustainable development according to their specific interests and situation. A multiregion approach to global sustainable development would be based on the sustainable provision of natural resources and the sustainable import and export of resources, goods, services and waste (Redclift, 1994).

A multi-region approach to global sustainable development would entail the elaboration of multiple concrete manifestations of sustainable development, that is, each region would be able to craft a distinct take on the sustainable development *problématique*. In order to give justice to the cultural, social, economic and ecological diversity of the world, multiple ways of interpreting sustainable development must be encouraged. Unfortunately, the powerful homogenization elements<sup>25</sup> inherent to the globalization process as currently unfolding conspire against those aspirations.

<sup>&</sup>lt;sup>25</sup> Particularly regarding national and international economic policies and reduction in the regulatory role of the State.

#### A variety of perspectives e)

Table I presents a summary of theoretical perspectives used to characterize sustainable development. It is important to recognize that sustainable development can be treated as both a model and a point of legitimation (Farrell and Hart, 1998). Accordingly, the term sustainable development is often used for different purposes in scientific and political realms (Drummond and Marsden, 1999). In the final analysis, no single group has authority to define sustainable development. Consequently, the concept is wed to ambiguity. This ambiguous character exemplifies the inherent rationalism of sustainable development (Drummond and Marsden, 1999).

Table 1

| Theory                         | Characterization of sustainable development   |
|--------------------------------|---|
| Equilibrium-<br>Neoclassical   | Welfare non-decreasing (anthropocentric); sustainable growth based on technology and<br>substitution; optimizing environmental externalities; maintaining the aggregate stock of natural and<br>economic capital; individual objectives prevail over social goals; policy needed when individual<br>objectives conflict; long-run policy based on market solutions.   |
| Neo-Austrian-<br>Temporal      | Teleological sequence of conscious and goal-oriented adaptation; preventing irreversible patterns; maintaining organization level (negentropy) in economic system; optimizing dynamic processes of extraction, production, consumption, recycling and waste treatment.  |
| Ecological-<br>Evolutionary    | Maintaining resilience of natural systems, allowing for fluctuation and cycles (regular destruction); learning from uncertainty in natural processes; no domination of food chains by humans; fostering genetic/biotic/ecosystem diversity; balanced nutrient flows in ecosystems.  |
| Evolutionary-<br>Technological | Maintaining co-evolutionary adaptive capacity in terms of knowledge and technology to react to uncertainties; fostering economic diversity of actors, sectors and technologies.   |
| Physico-<br>Economic           | Restrictions on materials and energy flows in/out the economy; industrial metabolism based on materials – product chain policy: integrated waste treatment, abatement, recycling and product development.   |
| Biophysical-<br>Energy         | A steady state with minimum materials and energy throughput; maintaining physical and biological stocks and biodiversity; transition to energy systems with minimum pollutive effects.  |
| Systems-<br>Ecological         | Controlling direct and indirect human effects on ecosystems; balance between material inputs and outputs to human systems; minimum stress factors on ecosystems, both local and global.   |
| Ecological<br>Engineering      | Integration of human benefits and environmental quality and functions by manipulation of ecosystems; design and improvement of engineering solutions on the boundary of economics, technology and ecosystems; utilizing resilience, self-organization, self-regulation and functions of natural systems for human purposes.   |
| Human<br>Ecology               | Remain within the carrying capacity (logistic growth); limited scale of economy and population; consumption oriented toward basic needs; occupy a modest place within the ecosystem food web and biosphere; always consider multiplier effects of human actions, in space and time.   |
| Socio-<br>Biological           | Maintain cultural and social system of interactions with ecosystems; respect for nature integrated in culture; survival of group important.   |
| Historical-<br>Institutional   | Equal attention to interests of nature, sectors and future generations; integrating institutional arrangements for economic and environmental policy; creating institutional long-run support for nature's interests; holistic instead of partial solutions, based on a hierarchy of values.  |
| Ethical-<br>Utopian            | New individual value systems (respect for nature and future generations, basic needs fulfillment) and new social objectives (steady state); balance attention for efficiency, distribution and scale; strive for small-scale activities and control of 'side effects' ('small is beautiful'); long-run policy based on changing values and encouraging citizen (altruistic) as opposed to individual (egoistic) behavior. |

#### THEORETICAL PERSPECTIVES ON SUSTAINABLE DEVELOPMENT

Source: Bergh and Jeroen (1996).

## VI. The different guises of development<sup>26</sup>

Development is about increasing the quality of life of human beings,<sup>27</sup> and not necessarily about increased GNP (not even of "greened" GNP which accounts for depletion of ecological capital).

Hence development is not synonymous with economic growth; the latter is only (one of) the means to the former. Development, in the last instance, is about improvements in the quality of life.

Quality of life embodies the satisfaction of material and nonmaterial human needs (resulting in the level of health reached) and the fulfillment of human desires and aspirations (resulting in the level of subjective satisfaction obtained). Human needs, desires and aspirations can be met through a variety of alternative material and non-material satisfiers (Maslow and Lowery, 1998, Gallopín and Öberg, 1992; Mallmann, 1980).

Contrary to common perception, economic growth is not necessarily synonymous with material growth. Material economic growth is now confronting both *source limitations* (scarcity of natural resources) and *sink limitations* (saturation of the natural capacity for dilution and neutralization of pollutants and wastes). Non-material economic growth has been increasing in the recent past; this relative dematerialization of the economy is evident in the increasing share of the services sector in the GNP (although not all services are immaterial, many are much less material-intensive than the

<sup>&</sup>lt;sup>26</sup> Gallopín (1996a), Gallopín and Christianson (2000).

<sup>&</sup>lt;sup>27</sup> Or, in economic terms, maximizing aggregate human welfare.

agricultural and industrial sectors of the economy) and the higher energy and resource efficiency of the new and emerging knowledge-intensive technologies.

Figure 8 represents the basic relation between development, economic growth, and material economic growth in the form of a Venn diagram familiar in set theory. Sustainability, in principle, increases along the axis material economic growth –non-material economic growth– no economic growth.<sup>28</sup> The figure is useful for mapping possible combinations of economic growth and changes in the quality of life (Gallopín, 1996a).

Figure 8 A SET-THEORETICAL REPRESENTATION OF DEVELOPMENT, SUSTAINABILITY, AND ECONOMIC GROWTH, AND QUALITY OF LIFE (QOFL)



Source: Author's elaboration.

One could say that *underdevelopment* occurs when neither quality of life increases nor economic growth takes place, a situation that affected many Latin American countries during the eighties and continues to plague many countries today, mostly in the south.

The situation where there is material economic growth, but quality of life does not increase, can be defined as *maldevelopment*; it occurs both in the north and in the south.

The combination of non-development with non-material economic growth is rare. However, it could characterize the situation of some fiscal havens or countries with service-based economies whose populations, for the most part, are resigned to a stagnant quality of life.

The combination of increasing quality of life with material economic growth is what is usually viewed as *development*. It currently occurs mostly in the north, but also in some countries

<sup>&</sup>lt;sup>28</sup> "No economic growth" can be consistent with development in the form of qualitative transformations.

in the south. However, in the long-term this situation is environmentally unsustainable, and in some instances (i.e., global climate change) critical environmental thresholds may have already been surpassed.

On our finite planet, even allowing for rapid technological change, a basic sustainable level of per capita material consumption will have to be reached. A reasonable way to do this will involve both increasing the material consumption of the billions of people living now in poverty and reducing material over-consumption by the rich minority.<sup>29</sup> Similarly, the global population will have to stabilize eventually.<sup>30</sup>

In the very long-term, there are two basic types of truly sustainable development situations: increasing quality of life with non-material economic growth (but no net material economic growth) and zero-growth economies (no economic growth at all). Sustainable development needs not imply the cessation of economic growth: a zero-growth material economy with a positively-growing non-material economy is the logical implication of sustainable development. While demographic growth and material economic growth must eventually stabilize, cultural, psychological and spiritual growth is not constrained by physical limits. Those situations are represented in Figure 9.



#### Figure 9 THE DIFFERENT GUISES OF DEVELOPMENT

Source: Author's elaboration.

<sup>&</sup>lt;sup>29</sup> Reducing material over-consumption by the rich minority can be achieved by reducing individual material consumption levels and/or by increasing the overall material and energetic efficiency of the economy.

<sup>&</sup>lt;sup>30</sup> Global population stabilization can be achieved through improving peoples living conditions and their quality of life. Stabilization through imposition and violence, besides being ineffectual, is unlikely to be conducive to sustainable development.

Rich countries should attempt to move from maldevelopment or development with material economic growth, to development with non-material economic growth (or, if society so desires, the transition could be to a zero-growth economy).

Underdeveloped countries in most cases, however, will be unable to move from nondevelopment to development without material economic-growth or to zero-growth economies, because of the fact that some level of accumulation and material economic activity is required to sustain development. The path from underdevelopment to maldevelopment is possible, but obviously inappropriate. Nevertheless, many countries continue to try and follow it.

After considering the alternatives, the only path realistically appropriate for developing countries, if sustainable development is to be achieved, is the one that goes from underdevelopment to development with material economic growth, and then to development without material economic growth. The paths discussed above, and the alternatives available in principle to rich and poor countries, are depicted in Figure 10.



## Figure 10 ALTERNATIVE DEVELOPMENT PATHS

Source: Author's elaboration.

## VII. Alternative paradigms for sustainable development

Schellnhuber (1998, 1999) developed an interesting perspective for sustainable development using a cybernetic approach in the context of global environmental change. Rather than attempting to formulate a single and binding definition, he proposes a set of precise optional paradigms<sup>31</sup> of coevolution of the human and natural subsystems of the (global) socio-ecological system, emphasizing different fundamental motives of human actions. The following discussion is an attempt to generalize his approach to any scale from the local to the global and to any socio-ecological system.

The state of a socio-ecological system can be represented by a point in a multidimensional "state space" defined by all possible values of the set of variables that define the ecological (or "natural") subsystem and the human subsystem. As the system state changes through time, the succession of states defines a trajectory for the system in this abstract state space.

For simplicity, in what follows all the variables describing the ecological subsystem and all those describing the human subsystem will be condensed down to only two variables, N and H respectively. N represents the state of the natural subsystem (for instance global mean temperature, or a local aggregate index of environmental

<sup>&</sup>lt;sup>31</sup> The term "paradigm" as used by Schellnhuber, seems to refer to a basic goal (such as optimization, stabilization, or other), a strategic direction, and the underlying assumptions about the functioning of the socio-ecological system.

conditions) and H represents the state of the human subsystem (e.g. the degree of development of human civilization, or the condition of the local human community).

Figure 11 shows the state space of the socio-ecological system. In this simplified example, the state space is the area defined by all possible values of the two variables N and H. There may be (depending of what are the specific variables represented by N) regions of the state space within which no life is possible (for instance, temperatures that are too cool or too hot to sustain human life). The values of N for which human life is possible constitute the ecological niche of humans.



#### Figure 11 AN IDEALIZED REPRESENTATION OF THE STATE SPACE OF A SOCIO-ECOLOGICAL SYSTEM

Source: Author's elaboration.

In the general case, the state space may contain "catastrophic domains", or regions where a) the quality of the socio-ecological system falls below a tolerable level, and b) once the state of the system enters the domain, it becomes trapped inside it. In the state space there may also exist 'inaccessible regions"; those are combinations of values of N and H that cannot be reached by any deliberate or spontaneous trajectory from the present or initial state  $P_0$ .

A generic state space for the coevolution of N and H will look like in Figure 12. Figure 13 illustrates some possible trajectories departing from the current state  $P_0$ .

#### Figure 12 A GENERIC STATE SPACE FOR REPRESENTING THE COEVOLUTION OF N AND H



Source: Author's elaboration.

#### a) Paradigm 1. Standardization

It implies the direct prescription of norms, values, development corridors, target domains, etc. for controlled coevolution of the N and H subsystems. In other words, it is based on the establishment of absolute environment and development standards, norms, quotas, or values to be reached. The coevolution criteria here are not strictly derived –or even derivable– from the internal dynamics of the socio-ecological system, but from essentially normative settings.

#### Figure 13 STATE TRAJECTORIES ORIGINATING FROM THE CURRENT STATE Po



Source: Author's elaboration.

This paradigm offers a seductively simple possibility for operationalizing sustainable development: certain environmental and human variables(often chosen to be as simple as possible) or entire aggregate functions are declared "sustainability indicators", and the trajectory of the socio-ecological system is considered correct if the values of the indicators are maintained within the limits defined as the "safe range".

The utilization of this paradigm implies the assumption that the considered socio-ecological system is steerable more or less "by sight", i.e. via the largely statistical, perpetual evaluation of space and time close-range information.

The fundamental arbitrariness in the stipulation of coevolution norms in this paradigm involves the great danger that essential interactions, repercussions and side effects in the non-linear socio-ecological systems to be controlled or steered are not taken into consideration.

Schellnhuber (1998) goes on to show that some management strategies which satisfies all desired standards over the short term can lead nevertheless to an irreversible development which destroys the long-term feasibility of the paradigm (the possibility of a trajectory being contained within the safe range).

#### b) Paradigm 2. Optimization

This implies a search for the "best", that is, maximizing an aggregated human-nature welfare function by choosing the optimal co-evolutionary trajectory over a fixed time period. Variations of this paradigm include the maximization of the mean utility over the long-term, or the acceptation of temporary drawbacks provided final utility is increased. Serious problems arise with this approach, including the analytical and political difficulties to define or impose target values for optimization. Another risk is that the success of the optimization paradigm rests on the assumption of perfect control over the socio-ecological system, but if that is not true, episodic departures from the ideal control scheme may result in the system falling in one of the catastrophic domains.

The optimization paradigm of sustainable development embodies an optimistic attitude that assumes that the best possible co-evolution can be actually realized under all circumstances. This requires rather perfect knowledge and a coherent volition process extending over many generations.

#### c) Paradigm 3. Pessimization

This paradigm (see also Gallopín 1997) aims to avoid falling into the catastrophic domains; it is based on the precautionary principle of "preventing the worst", looking for the smallest possible amount of damage instead of the greatest possible benefit. This paradigm does not primarily determine the pessimal management sequence, but rather tries to exclude non-tolerable control options and therefore, a fairy large maneuvering space for management option usually remains.

#### d) Paradigm 4. Equitization

This focuses on preserving the options for future generations; in other words, not contracting the "accessible universe" over time. "Equity" here is identified with equality of environment and development options for successive generations. To fulfill such a goal, a full analysis of all possible trajectories emanating from the present would be needed, and this is likely to be an impossible task, even in principle. Moreover, the dynamics of the socio-ecological system might not permit a trajectory fulfilling the preservation of *the same* options through time; the goal could then be to preserve *equivalent* options (accepting that some options could not be preserved but that new options will become possible).

The comparison of the options lost to the options gained to determine whether the overall opportunistic quality of the set of trajectories will be preserved is a most difficult task. Of course, it would be possible to introduce a short-term option criterion in the standardization paradigm for sustainable development on a short-term basis, but this will suffer from the same problems of that paradigm.

#### e) Paradigm 5. Stabilization

This focuses on bringing the socio-ecological system into a desirable state in the coevolution on state space and then maintaining it by good management. Note that the stabilization paradigm shifts the focus from "sustainable development" to "sustainability". The stabilization paradigm does not simply prescribe co-evolution states or paths, but searches systematically for balancing management options in accordance with the intrinsic dynamics of the socio-ecological system and the available repertoire of steering instruments. If the desired generalized equilibrium exists and is accessible, then criteria are needed for the choice of the trajectory leading to it (e.g. "soft-landing" or "crash-halt").

In the real world it is to be expected that, rather than pure paradigms, strategies trying to satisfy simultaneously a number of them would be implemented (e.g. complex paradigms representing a ranking or other combination of various pure paradigms).

The characterization of pure paradigms is nevertheless illuminating, not only because it highlights the fundamental dimensions involved, but also because it shows vividly that the challenge of choosing the "right actions" is not only normative (what society wants) but also ontological and epistemological (how and by which laws socio-ecological systems operate, and what do we know about them). This tri-layered complexification (volition, understanding, and socio-ecological dynamics) is typical of the problems of sustainable development (Gallopín et al., 2001).

## **VIII.** Conclusions

The major conclusions obtained from a systemic analysis of the concepts of sustainability and sustainable development are presented below.

When discussing sustainability, the clear specification of the system (or the outputs of the system) to which the concept is applied (What system? How defined? At what scale? Which outputs?) is essential to avoid confusion and ambiguity.

Many disagreements regarding the concrete meaning and implications of sustainability are associated to the use of different valuation criteria (or valuation function) utilized (i.e. relative weight allocated to natural and manufactured capital); therefore it is critical to clearly specify the criteria adopted.

Sustainability is a property of a system open to interactions with its external world. It is not a fixed state of constancy, but a dynamic preservation of the essential identity of the system amidst permanent change. A small number of generic attributes may provide the foundations of sustainability.

Sustainable development is not a property but a process of directional change by which a system improves<sup>32</sup> through time in a sustainable way.

Development and economic growth are often confounded, but

<sup>&</sup>lt;sup>32</sup> Improvement ("change for the better") is a normative concept, and thus the definition of when a change represents an improvement may differ among parties adopting different paradigms of sustainable development.

they need to be clearly differentiated; development is a qualitative process of realization of potentialities which may or may not involve economic growth (a quantitative increase in wealth).

Economic growth and matter/energy throughput must be decoupled: economic growth is not necessarily synonymous with material economic growth.

Different situations and strategies regarding sustainability of development can be categorized along the dimensions of quality of life, material economic growth, and non-material economic growth.

#### References

- Ashby, W.R (1956), An Introduction to Cybernetics. Chapman & Hall Ltd., London.
- Ayres R.U., J.C.J.M. van den Bergh and J.M. Gowdy (1998), Viewpoint: Weak Versus Strong Sustainability. Tinbergen Institute Discussion Papers 98-103/3 / Tinbergen Institute (web site) (RePEc:dgr:uvatin:19980103); http://netec.ier.hit-u.ac.jp/WoPEc/data/Papers/dgruvatin19980103.html
- Bergh, van den, and C.J.M. Jeroen (1996), "Sustainable Development and Management", *Ecological Economics and Sustainable Development: Theory, Methods and Applications*, pp. 53-79, Cheltenham, U.K., Edward Elgar Publishing.
- Brooks, Harvey (1992), Sustainability and Technology, Science and Sustainability: Selected Papers on IIASA's 25<sup>th</sup> Anniversary, pp. 29-60, Vienna, IIASA.
- Castri, F. di (1981), La ecología moderna: génesis de una ciencia del hombre y de la naturaleza. *El Correo de la UNESCO*, abril 1981, pp. 6-11.
- Costanza, Robert (1991), The Ecological Economics of Sustainability: Investing in Natural Capital, *Environmentally Sustainable Economic Development: Building on Brundtland*, R. Goodland, H. Daly, S. El Serafy and B. von Droste (eds.), pp. 83-90, New York, UNESCO.
- Drengson, A. and Y. Inoue (Editors) (1995). *The Deep Ecology Movement: An Introductory Anthology*. North Atlantic Books, N.Y.
- Drummond, Ian and Terry Marsden (1999), Sustainable Development: The Impasse and Beyond, *The Condition of Sustainability*, pp. 7-23, London, Routledge.
- Farrell, Alex and Maureen Hart (1998), What Does Sustainability Really Mean? –The Search for Useful Indicators, *Environment* 40 (9): 4-9 and 26-31.
- Froger, G. and E. Zyla (1998), Towards a Decision–Making Framework to Address Sustainable Development Issues, Sustainable Development: Concepts, Rationalities and Strategies, Sylvie Faucheux, Martin O'Connor

and Jan van der Straaten (eds.), London, Kluwer Academic Publishers.

- Gallopín, G.C. (1994), *Impoverishment and Sustainable Development. A Systems Approach*. International Institute for Sustainable Development; Winnipeg, Canada.
- Gallopín, G.C. (1996), Environmental and Sustainability Indicators and the Concept of Situational Indicators. A Systems Approach, *Environmental Modelling & Assessment* 1(3): 101-117.
- Gallopín, G.C. (1997), Branching Futures and Energy Projections. *Renewable Energy for Development* 10 (3): 4-5.
- Gallopín, G.C. and K. Christianson (2000), Sustainable Development, Society and the Environment: A Conceptual Framework for Tracking the Linkages, Stockholm Environment Institute, Stockholm.
- Gallopín G.C., P. Gutman, and H. Maletta, (1989), Global Impoverishment, Sustainable Development and the Environment: A Conceptual Approach, *ISSJ* 121: 375 397.
- Gallopín, G.C., S. Funtowicz, M. O'Connor, and J. Ravetz (2001), Science for the 21st Century: from Social Contract to the Scientific Core. *Int. Journal Social Science* 168: 219-229.
- Gill, A. (1969), *Finite-State Systems*, pp. 43-94, L.A. Zadeh and E. Polak (eds.), "System Theory", McGraw-Hill, New York.
- Gunderson, L.H., C.S. Holling and S.S. Light (eds.) (1995), *Barriers and Bridges to the Renewal of Ecosystems and Institutions*, Columbia University Press, New York.
- Holling, C.S. (1973), Resilience and Stability of Ecological Systems, Ann. Rev. Ecol. & Systematics 4: 1-23.
- Holling, C.S. (1986), The Resilience of Terrestrial Ecosystems: Local Surprise and Global Change, W.C. Clark & R.E. Munn (eds.), *Sustainable Development of the Biosphere*, IIASA/Cambridge University Press, pp. 292-317, Cambridge.
- IGBP (International Geosphere-Biosphere Programme) (2001), *Global Change and the Earth System: A planet under pressure*. The Global Environmental Programmes. Edited by Will Steffen and Peter Tyson. Stockholm: IGBP, IGBP Science Series No. 4, 32pp (available at http://www.igbp.kva.se/cgi-bin/php/frameset.php).
- Kates, R.W., W.C. Clark, R. Corell, J. M. Hall, C.C. Jaeger, I. Lowe, J.J. McCarthy, H.J. Schellnhuber, B. Bolin, N.M. Dickson, S. Faucheux, G.C. Gallopin, A. Gruebler, B. Huntley, J. Jäger, N.S. Jodha, R.E. Kasperson, A. Mabogunje, P. Matson, H. Mooney, B. Moore III, T. O'Riordan, and U. Svedin. (2001) Sustainability science. Science 292: 641-642.
- Maslow, A., & Lowery, R. (Ed.). (1998). Toward a psychology of being (3rd ed.). New York: Wiley & Sons.
- Nicolis, G. and I. Prigogine (1977), Self-Organization in Non-Equilibrium Systems: From Dissipative Structures to Order Through Fluctuations, New York, Wiley.
- Pearce, D. W. & Atkinson, G. (1992), Are national economies sustainable? Measuring sustainable development. CSERGE GEC Working Paper 92-11. University College London. p. 18.
- Peterson, Tarla Rai (1997), *Sustainable Development Comes of Age*, Sharing the Earth: The Rhetoric of Sustainable Development, pp. 6-33, Columbia, South Carolina, University of South Carolina Press.
- Prigogine, I. et I. Stengers (1979), La Nouvelle Alliance. Métamorphose de la Science, Gallimard, Paris.
- Redclift, Michael (1994), Sustainable Development: Economics and the Environment, Strategies for Sustainable Development: Local Agendas for the Southern Hemisphere (eds.), Redclift, Michael and Colin Sage, pp. 17-34, New York, John Wiley and Sons.
- Schullnhuber, H.J. (1998). Earth System Analysis. The Scope of the Challenge; p. 3-195 in Schullnhuber, H.J. and V. Wenzel (Eds.) 1998. Earth System Analysis. Integrating Science for Sustainability. Springer, Berlin.
- Schullnhuber, H.J. (1999). 'Earth system' analysis and the second Copernican revolution. *Nature* 402: c19-C23.
- Turner, R. Kerry (1993), Sustainability: principles and practice. In: R. Kerry Turner (ed.). Sustainable environmental economics and management: principles and practise. Belhaven Press, London. p. 3-36.
- UNCED (1992), Agenda 21. United Nations Conference on Environment and Development (UNCED); United Nations.
- Vercelli, Alessandro (1998), "Sustainable Development and the Freedom of Future Generations", Sustainability: Dynamics and Uncertainty, Chichilnisky, Graciela, Geoffrey M. Heal and Alessandro Vercelli (eds.), London, Kluwer Academic Publishers.

Wilson, Edward, O. (1986), Biophilia, Cambridge, Massachusetts, Harvard University Press.

World Commission on Environment and Development (WCED) (1987). *Our Common Future*; Oxford. Oxford University Press.



## CEPALSerieECLACmedio ambiente y desarrollo

#### **Published numbers**

- 1. Las reformas del sector energético en América Latina y el Caribe (LC/L.1020), abril de 1997. E-mail: <u>fsanchez@eclac.cl</u> <u>haltomonte@eclac.cl</u>
- 2. Private participation in the provision of water services. Alternative means for private participation in the provision of water services (LC/L.1024), May 1997. E-mail: <u>ajoravlev@eclac.cl</u>
- 3. Management procedures for sustainable development (applicable to municipalities, micro region and river basins) (LC/L.1053), August 1997. E-mail: <u>adourojeanni@eclac.cl</u>, <u>rsalgado@eclac.cl</u>
- 4. El Acuerdo de las Naciones Unidas sobre pesca en alta mar: una perspectiva regional a dos años de su firma (LC/L.1069), septiembre de 1997. E-mail: <u>rsalgado@eclac.cl</u>
- 5. Litigios pesqueros en América Latina (LC/L.1094), febrero de 1998. E-mail: rsalgado@eclac.cl
- 6. Prices, property and markets in water allocation (LC/L1097), February 1998. E-mail: <u>tlee@eclac.cl</u> <u>ajouralev@eclac.cl</u>

Los precios, la propiedad y los mercados en la asignación del agua (LC/L.1097), octubre de 1998. E-mail: tlee@eclac.cl - ajouralev@eclac.cl

- 7. Sustainable development of human settlements: Achievements and challenges in housing and urban policy in Latin America and the Caribbean (LC/L.1106), March 1998. E-mail: <u>dsimioni@eclac.cl</u> www Desarrollo sustentable de los asentamientos humanos: Logros y desafíos de las políticas habitacionales y urbanas de América Latina y el Caribe (LC/L.1106), octubre de 1998. <u>dsimioni@eclac.cl</u> www
- Hacia un cambio de los patrones de producción: Segunda Reunión Regional para la Aplicación del Convenio de Basilea en América Latina y el Caribe (LC/L.1116 y LC/L.1116 Add/1) vols. I y II, en edición. E-mail: <u>cartigas@eclac.cl</u> - <u>rsalgados@eclac.cl</u>
- 9. La industria del gas natural y las modalidades de regulación en América Latina, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1121), abril de 1998. E-mail <u>fsanchez@eclac.cl</u> www
- **10.** Guía para la formulación de los marcos regulatorios, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1142), agosto de 1998. E-mail: <u>fsanchez@eclac.cl</u> www
- Panorama minero de América Latina: la inversión en la década de los noventa, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1148), octubre de 1998. E-mail: <u>fsanchez@eclac.cl</u> www
- 12. Las reformas energéticas y el uso eficiente de la energía en el Perú, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1159), noviembre de 1998. E-mail: <u>fsanchez@eclac.cl</u> www
- **13.** Financiamiento y regulación de las fuentes de energía nuevas y renovables: el caso de la geotermia (LC/L.1162) diciembre de 1998. E-mail: mcoviello@eclac.cl www
- 14. Las debilidades del marco regulatorio eléctrico en materia de los derechos del consumidor. Identificación de problemas y recomendaciones de política, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1164), enero de 1999. E-mail: <u>fsanchez@eclac.cl</u> www
- 15. Primer Diálogo Europa-América Latina para la Promoción del Uso Eficiente de la Energía, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1187), marzo de 1999. E-mail: <u>fsanchez@eclac.cl</u> www
- 16. Lineamientos para la regulación del uso eficiente de la energía en Argentina, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1189), marzo de 1999. E-mail: <u>fsanchez@eclac.cl</u> www
- 17. Marco legal e institucional para promover el uso eficiente de la energía en Venezuela, Proyecto CEPAL/Comisión Europea "Promoción del uso eficiente de la energía en América Latina" (LC/L.1202), abril de 1999. E-mail: <u>fsanchez@eclac.cl</u> www

- **18.** Políticas e instituciones para el desarrollo sostenible en América Latina y el Caribe, José Antonio Ocampo (LC/L.1260-P), N° de venta: S.99.II.G.37 (US\$10.00), septiembre de 1999. E-mail: jocampo@eclac.cl
- Impactos ambientales de los cambios en la estructura exportadora en nueve países de América Latina y el Caribe: 1980-1995, Marianne Schaper (LC/L.1241/Rev1-P), Nº de venta: S.99.II.G.44 (US\$10.00), octubre de 2000. E-mail: mschaper@eclac.cl www
- 20. Marcos regulatorios e institucionales ambientales de América Latina y el Caribe en el contexto del proceso de reformas macroeconómicas: 1980-1990, Guillermo Acuña (LC/L.1311-P), Nº de venta: S.99.II.G.26 (US\$10.00), diciembre de 1999. E-mail: gacuna@eclac.cl www
- 21. Consensos urbanos. Aportes del Plan de Acción Regional de América Latina y el Caribe sobre Asentamientos Humanos, Joan MacDonald y Daniela Simioni (LC/L.1330-P), Nº de venta: S.00.II.G.38 (US\$10.00), diciembre de 1999. E-mail: <u>dsimioni@eclac.cl</u> www Urban consensus. Contributions from the Latin America and the Caribbean Regional Plan of Action on Human Settlements, Joan MacDonald y Daniela Simioni (LC/L.1330-P), Sales Nº: E.00.II.G.38 (US\$10.00), June 2000. E-mail: <u>dsimioni@eclac.cl</u> www
- 22. Contaminación industrial en los países latinoamericanos pre y post reformas económicas, Claudia Schatan (LC/L.1331-P), Nº de venta: S.00.II.G.46 (US\$10.00), diciembre de 1999. E-mail: mschaper@eclac.cl
- Trade liberation and industrial pollution in Brazil, Claudio Ferraz and Carlos E.F. Young (LC/L.1332-P), Sales N°: E.00.II.G.47 (US\$10.00), December 1999. E-mail: <u>mschaper@eclac.cl</u> www
- 24. Reformas estructurales y composición de las emisiones contaminantes industriales. Resultados para México, Fidel Aroche Reyes (LC/L.1333-P), Nº de venta: S.00.II.G.42 (US\$10.00), mayo de 2000. E-mail: mschaper@eclac.cl www
- 25. El impacto del programa de estabilización y las reformas estructurales sobre el desempeño ambiental de la minería de cobre en el Perú: 1990-1997, Alberto Pascó-Font (LC/L.1334-P), Nº de venta: S.00.II.G.43, (US\$10.00), mayo de 2000. E-mail: mschaper@eclac.cl www
- 26. Servicios urbanos y equidad en América Latina. Un panorama con base en algunos casos, Pedro Pírez (LC/L.1320-P), Nº de venta: S.00.II.G.95, (US\$10.00), septiembre de 2000. E-mail: <u>dsimioni@eclac.cl</u> www
- 27. Pobreza en América Latina: Nuevos escenarios y desafíos de políticas para el hábitat urbano, Camilo Arraigada (LC/L.1429-P), Nº de venta: S.00.II.G.107, (US\$10.00), octubre de 2000. E-mail: dsimioni@eclac.cl www
- Informalidad y segregación urbana en América Latina. Una aproximación, Nora Clichevsky (LC/L.1430-P), N° de venta: S.99.II.G.109, (US\$10.00), octubre de 2000. E-mail: <u>dsimioni@eclac.cl</u> www
- **29.** Lugares o flujos centrales: los centros históricos urbanos, Fernando Carrión (LC/L.1465-P), N° de venta: S.01.II.G.6, (US\$10.00), diciembre de 2000. E-mail: <u>rjordan@eclac.cl</u> www
- 30. Indicadores de gestión urbana. Los observatorios urbano-territoriales para el desarrollo sostenible. Manizales, Colombia, Luz Stella Velásquez (LC/L.1483-P), Nº de venta: S.01.II.G.24, (US\$10.00), enero de 2001. E-mail: rjordan@eclac.cl www
- **31.** Aplicación de instrumentos económicos en la gestión ambiental en América Latina y el Caribe: desafíos y factores condicionantes, Jean Acquatella (LC/L.1488-P), Nº de venta: S.01.II.G.28, (US\$10.00), enero de 2001. E-mail: jacquatella@eclac.cl www
- 32. Contaminación atmosférica y conciencia ciudadana. El caso de la ciudad de Santiago, Cecilia Dooner, Constanza Parra y Cecilia Montero (LC/L.1532-P), Nº de venta: S.01.II.G.77, (US\$10.00), abril de 2001. Email: <u>dsimioni@eclac.cl</u> www
- **33.** Gestión urbana: plan de descentralización del municipio de Quilmes, Buenos Aires, Argentina, Eduardo Reese (LC/L.1533-P), Nº de venta: S.01.II.G.78, (US\$10.00), abril de 2001. E-mail: <u>rjordan@eclac.cl</u> www
- **34.** Gestión urbana y gobierno de áreas metropolitanas, Alfredo Rodríguez y Enrique Oviedo (LC/L.1534-P), N° de venta: S.01.II.G.79, (US\$10.00), mayo de 2001. E-mail: <u>rjordan@eclac.cl</u> www
- 35. Gestión urbana: recuperación del centro de San Salvador, El Salvador. Proyecto Calle Arce, Jaime Barba y Alma Córdoba (LC/L.1537-P), Nº de venta: S.01.II.G.81, (US\$10.00), mayo de 2001. E-mail: rjordan@eclac.cl www
- 36. Consçiêcia dos cidadãos o poluição atmosférica na região metropolitana de São Paulo RMSP, Pedro Roberto Jacobi y Laura Valente de Macedo (LC/L.1543-P), Nº de venta: S.01.II.G.84, (US\$10.00), mayo de 2001. E-mail: <u>dsimioni@eclac.cl</u> www
- Environmental values, valuation methods, and natural damage assessment, Cesare Dosi (LC/L.1552-P), Sales N°: E.01.II.G.93, (US\$10.00), June 2001. E-mail: <u>dsimioni@eclac.cl</u> www
- 38. Fundamentos económicos de mecanismos de flexibilidad para la reducción internacional de emisiones en el marco de la Convención de cambio Climático (UNFCCC), Jean Acquatella (LC/L.1556-P), Nº de venta: S.01.II.G.101, (US\$10.00), julio de 2001. E-mail: jacquatella@eclac.cl www

- **39.** Fundamentos territoriales y biorregionales de la planificación, Roberto Guimarães (LC/L.1562-P), Nº de venta: S.01.II.G.108, (US\$10.00), julio de 2001. E-mail: <u>rguimaraes@eclac.cl</u> www
- 40. La gestión local, su administración, desafíos y opciones para el fortalecimiento productivo municipal en Caranavi, Departamento de La Paz, Bolivia, Jorge Salinas (LC/L.1577-P), Nº de venta: S.01.II.G.119, (US\$10.00), agosto de 2001. E-mail: jsalinas@eclac.cl www
- 41. Evaluación ambiental de los acuerdos comerciales: un análisis necesario, Carlos de Miguel y Georgina Núñez (LC/L.1580-P), Nº de venta: S.01.II.G.123, (US\$10.00), agosto de 2001. E-mail: <u>cdemiguel@eclac.cl</u> y <u>gnunez@eclac.cl</u> www
- **42.** Nuevas experiencias de concentración público-privada: las corporaciones para el desarrollo local, Constanza Parra y Cecilia Dooner (LC/L.1581-P), N° de venta: S.01.II.G.124, (US\$10.00), agosto de 2001. E-mail: rjordan@eclac.cl www
- 43. Organismos genéticamente modificados: su impacto socioeconómico en la agricultura de los países de la Comunidad Andina, Mercosur y Chile, Marianne Schaper y Soledad Parada (LC/L.1638-P), Nº de venta: S.01.II.G.176, (US\$10.00), noviembre de 2001. E-mail: mschaper@eclac.cl www
- **44.** Dinámica de valorización del suelo en el área metropolitana del Gran Santiago y desafíos del financiamiento urbano, Camilo Arraigada Luco y Daniela Simioni (LC/L.1646-P), Nº de venta: S.01.II.G.185, (US\$10.00), noviembre de 2001. E-mail: <u>dsimioni@eclac.cl</u> www
- 45. El ordenamiento territorial como opción de políticas urbanas y regionales en América Latina y el Caribe, Pedro Felipe Montes Lira (LC/L.1647-P), Nº de venta: S.01.II.G.186, (US\$10.00), diciembre de 2001. E-mail: rjordan@eclac.cl www
- 46. Evolución del comercio y de las inversiones extranjeras e industrias ambientalmente sensibles: Comunidad Andina, Mercosur y Chile (1990-1999), Marianne Schaper y Valerie Onffroy de Vèréz (LC/L.1676-P), Nº de venta: S.01.II.G.212, (US\$10.00), diciembre de 2001. E-mail: <u>mschaper@eclac.cl</u> www
- 47. Aplicación del principio contaminador-pagador en América Latina. Evaluación de la efectividad ambiental y eficiencia económica de la tasa por contaminación hídrica en el sector industrial colombiano, Luis Fernando Castro, Juan Carlos Caicedo, Andrea Jaramillo y Liana Morera (LC/L.1691-P), Nº de venta: S.02.II.G.15, (US\$10.00), febrero de 2002. E-mail: jacquatella@eclac.cl www
- **48.** Las nuevas funciones urbanas: gestión para la ciudad sostenible, (varios autores) (LC/L.1692-P), Nº de venta: S.02.II.G.32, (US\$10.00), abril de 2002. E-mail: <u>dsimioni@eclac.cl</u> www
- Pobreza y políticas urbano-ambientales en Argentina, Nora Clichevsky (LC/L.1720-P), N
  <sup>o</sup> de venta: S.02.II.G.31, (US\$10.00), abril de 2002. E-mail: <u>dsimioni@eclac.cl</u> www
- **50.** Políticas públicas para la reducción de la vulnerabilidad frente a los desastres naturales, Jorge Enrique Vargas (LC/L.1723-P), Nº de venta: S.02.II.G.34, (US\$10.00), abril de 2002. E-mail: <u>dsimioni@eclac.cl</u> www
- 51. Uso de instrumentos económicos para la gestión ambiental en Costa Rica, Jeffrey Orozco B. y Keynor Ruiz M. (LC/L.1735-P), Nº de venta: S.02.II.G.45, (US\$10.00), junio de 2002. E-mail: jacquatella@eclac.cl www
- 52. Gasto, inversión y financiamiento para el desarrollo sostenible en Argentina, Daniel Chudnovsky y Andrés López (LC/L.1758-P), Nº de venta: S.02.II.G.70, (US\$10.00), octubre de 2002. E-mail: <u>cdemiguel@eclac.cl</u> www
- 53. Gasto, inversión y financiamiento para el desarrollo sostenible en Costa Rica, Gerardo Barrantes (LC/L.1760-P), Nº de venta: S.02.II.G.74, (US\$10.00), octubre de 2002. E-mail: <u>cdemiguel@eclac.cl</u> www
- 54. Gasto, inversión y financiamiento para el desarrollo sostenible en Colombia, Francisco Alberto Galán y Francisco Javier Canal (LC/L.1788-P) N° de venta: S.02.II.G.102 (US\$ 10.00), noviembre de 2002. E-mail: cdemiguel@eclac.cl www
- 55. Gasto, inversión y financiamiento para el desarrollo sostenible en México, Gustavo Merino y Ramiro Tovar (LC/L.1809-P) N° de venta: S.02.II.G.102 (US\$ 10.00), noviembre de 2002. E-mail: <u>cdemiguel@eclac.cl</u>
- 56. Expenditures, Investment and Financing for Sustainable Development in Trinidad and Tobago, Desmond Dougall and Wayne Huggins (LC/L.1795-P), Sales N°: E.02.II.G.107 (US\$ 10.00), November, 2002. E-mail: cdemiguel@eclac.cl www
- Gasto, inversión y financiamiento para el desarrollo sostenible en Chile, Francisco Brzovic (LC/L.1796-P), N° de venta: S.02.II.G.108 (US\$ 10.00), noviembre de 2002. E-mail: <u>cdemiguel@eclac.cl</u> www
- 58. Expenditures, Investment and Financing for Sustainable Development in Trinidad and Tobago, Desmond Dougall and Wayne Huggins (LC/L.1795-P), Sales N°: E.02.II.G.107 (US\$ 10.00), November, 2002. E-mail: <a href="mailto:cdemiguel@eclac.cl">cdemiguel@eclac.cl</a> www
- **59.** En Prensa.
- **60.** En Prensa.
- **61.** En Prensa.
- **62.** En Prensa.

- **63.** Necesidades de bienes y servicios para el mejoramiento ambiental de la PYME en Chile. Identificación de factures críticos y diagnóstico del sector, José Leal (LC/L.1851-P), N° de venta: S.03.II.G.15 (US\$ 10.00), marzo de 2003. E-mail: <u>mschaper@eclac.cl</u> www
- **64.** A systems approach to sustainability and sustainable development, Gilberto Gallopín (LC/L.1864-P), Sales N°: E.03.II.G.35 (US\$ 10.00), March 2003. E-mail: <u>ggallopin@eclac.cl</u> www

| www: These publications are also | available on the internet: <u>http://www.eclac.cl</u> |
|----------------------------------|---|
|----------------------------------|---|

| Name:                        |
|------------------------------|
| Activity:                    |
| Address:                     |
| Postal, code, city, country: |
| Tel.:                        |
|                              |

 $_{\rm X}$  Readers wishing to obtain the above publications can do so by writing to the following address: ECLAC, Sustainable Development and Human Settlements Division, Distribution Unit, Casilla 179-D, Santiago, Chile.

x Publications available for sale should be ordered from the Distribution Unit, ECLAC, Casilla 179-D, Santiago, Chile, Fax (562) 210-2069, publications@eclac.cl.